

Dear editor,

We would like to thank you for your invitation to contribute an article to your high impact journal. We have revised the article according to your journal guidelines and the reviewers' comments.

Reviewer 1:

Thank you for your kind feedback. A table has been added (table 4) with the advantages and disadvantages of each cooperative technique.

Sincerely,

Dimitrios Ntourakis

Athens, 14/9/2015